Co-occurrence of Point Mutations in the Voltage-Gated Sodium Channel of Pyrethroid-Resistant *Aedes aegypti* Populations in Myanmar



Hitoshi Kawada¹*, Sai Zaw Min Oo^{1,2}, Sein Thaung², Emiko Kawashima¹, Yan Naung Maung Maung², Hlaing Myat Thu², Kyaw Zin Thant², Noboru Minakawa^{1,3}

1 Department of Vector Ecology and Environment, Institute of Tropical Medicine, Nagasaki University, Nagasaki, Japan, 2 Medical Entomology Research Division, Department of Medical Research (Lower Myanmar), Yangon, Myanmar, 3 The Global Center of Excellence Program, Nagasaki University, Nagasaki, Japan

Abstract

Background: Single amino acid substitutions in the voltage-gated sodium channel associated with pyrethroid resistance constitute one of the main causative factors of knockdown resistance in insects. The *kdr* gene has been observed in several mosquito species; however, point mutations in the para gene of *Aedes aegypti* populations in Myanmar have not been fully characterized. The aim of the present study was to determine the types and frequencies of mutations in the para gene of *Aedes aegypti* collected from used tires in Yangon City, Myanmar.

Methodology/Principal Findings: We determined high pyrethroid resistance in *Aedes aegypti* larvae at all collection sites in Yangon City, by using a simplified knockdown bioassay. We showed that V1016G and S989P mutations were widely distributed, with high frequencies (84.4% and 78.8%, respectively). By contrast, we were unable to detect I1011M (or I1011V) or L1014F mutations. F1534C mutations were also widely distributed, but with a lower frequency than the V1016G mutation (21.2%). High percentage of co-occurrence of the homozygous V1016G/S989P mutations was detected (65.7%). Additionally, co-occurrence of homozygous V1016G/F1534C mutations (2.9%) and homozygous V1016G/F1534C/S989P mutations (0.98%) were detected in the present study.

Conclusions/Significance: Pyrethroid insecticides were first used for malaria control in 1992, and have since been constantly used in Myanmar. This intensive use may explain the strong selection pressure toward *Aedes aegypti*, because this mosquito is generally a domestic and endophagic species with a preference for indoor breeding. Extensive use of DDT for malaria control before the use of this chemical was banned may also explain the development of pyrethroid resistance in *Aedes aegypti*.

Citation: Kawada H, Oo SZM, Thaung S, Kawashima E, Maung YNM, et al. (2014) Co-occurrence of Point Mutations in the Voltage-Gated Sodium Channel of Pyrethroid-Resistant Aedes aegypti Populations in Myanmar. PLoS Negl Trop Dis 8(7): e3032. doi:10.1371/journal.pntd.0003032

Editor: Pattamaporn Kittayapong, Mahidol University, Thailand

Received March 3, 2014; Accepted May 29, 2014; Published July 31, 2014

Copyright: © 2014 Kawada et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability: The authors confirm that all data underlying the findings are fully available without restriction. The unique DNA haplotype sequences were deposited in GenBank (DDBJ, accession = AB914689, AB914690, AB914687, AB914688).

Funding: This study was funded by Japan Initiative for Global Research Network on Infectious Diseases. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* Email: vergiss@nagasaki-u.ac.jp

Introduction

Pyrethroids constitute a group of chemicals that are structurally modified by natural pyrethrum, and are composed of two main types. Most of the 1st generation pyrethroids are knockdown agents, which possess high knockdown activity but low killing activity; the second generation pyrethroids possess high killing activity. The use of knockdown agents such as *d*-allethrin and metofluthrin—which generally exhibit low stability in the environment—as "spatial repellents" is biorational, because it causes low selection pressure on insect populations, thereby minimizing the development of physiological resistance. On the other hand, killing agents such as permethrin, deltamethrin, cypermethrin, and λ -cyhalothrin generally exhibit high photostability, and this enables their outdoor use in agriculture and their predominant application as vector control agents. Globally, photostable pyrethroids comprise 40% of the insecticides used annually for indoor residual spraying against malaria vectors [1]. However, the high stability and killing efficacy of pyrethroids has accelerated the development of pyrethroid resistance in vector mosquitoes [2].

Resistance to photostable pyrethroids is a major problem for the vector control program. Pyrethroid resistance involves two main mechanisms, namely, enhanced metabolic detoxification and insensitivity of target sites. Single amino acid substitutions in the voltage-gated sodium channel, known as kdr mutations, constitute one of the main causative factors of pyrethroid resistance. This kdr-type resistance has been observed in several mosquitoes,

Author Summary

The use of pyrethroids with high killing activity has accelerated the development of pyrethroid resistance in vector mosquitoes. The knockdown resistance (kdr) allele contains a single amino acid substitution in the voltagegated sodium channel and is one of the main causative factors of pyrethroid resistance in insects. We investigated the distribution of the kdr gene in Aedes aegypti larvae collected from used tires in Yangon City, Myanmar. We detected three patterns of co-occurrence of point mutations, namely, V1016G/S989P with wide distribution, and small number of V1016G/F1534C and V1016G/F1534C/ S989P, both of which were first found in the field collected Ae. aegypti population. The use of pyrethroids and/or DDT for malaria control is thought to be one of the main causes of pyrethroid resistance in Aedes aegypti. Insecticide treatment for malaria vector control seems to have been intensively conducted in the interior and along the periphery of human habitation areas, where the breeding and resting sites of Aedes aegypti are located.

including Anopheles gambiae Giles [3], Anopheles stephensi Liston [4], Culex quinquefasciatus Say [5], and Aedes aegypti (L.) [6]. Several mutations in segment 6 of domain II of the voltage-gated sodium channel were reported to play important roles in pyrethroid resistance of Aedes aegypti (I1011M, I1011V, V1016G, and V1016I) [6-8]. Recently, Yanola et al. identified a novel F1534C mutation in segment 6 of domain III in DDT/ permethrin-resistant Aedes aegypti [9,10]. Harris et al. reported that the F1534C mutation was strongly correlated with resistance to DDT and pyrethroid [11]. The S989P mutation in domain II of the voltage-gated sodium channel gene, which occurs in deltamethrin-resistant Aedes aegypti, is another principal kdr mutation that regulates pyrethroid resistance in mosquitoes [12]. Yanola et al. [10] reported the existence of the F1534C mutation in Aedes aegypti from Yangon City, Myanmar. However, to the best of our knowledge, no other detailed genetic analyses of point mutations in the voltage-gated sodium channel of Aedes aegypti colonies in Myanmar have been performed.

The objective of the present study was to elucidate the mechanisms of pyrethroid resistance in *Aedes aegypti* collected from used tires in Yangon City, Myanmar, by analyzing the presence of mutations in the voltage-gated sodium channel gene. Our targets were the four most frequent amino acid replacements in S989, I1011, L1014, and V1016—all of which are located in the area of segment 6 of domain II—and also a recently identified amino acid replacement at F1534—located in the area of segment 6 of domain III.

Methods

Collection of *Aedes aegypti* larvae from used tires and application of a simplified knockdown bioassay

We drove along the main roads in five urban townships (Tharketa, Thingangyun, North Dagon, South Okklapa, and Tarmwe) and two suburban townships (Thanlyin and Hlegu) in Yangon City, from September 26 to October 4, 2013, and collected mosquito larvae from used tires. Whenever we encountered used tires, most of which were found along the periphery of repair shops, we recorded the geographical location of the collection site by using a global positioning system (GPS). In addition, we took photographs of the surrounding area to record environmental characteristics such as vegetation, urbanization, etc. We collected mosquito larvae from the tires with netting. We performed a simplified bioassay for the detection of knockdown susceptibility in the Aedes larvae on the day of collection, according to Kawada et al. [13]. Each larva was placed in a glass vial containing 20 mL of water. An emulsifiable concentrate of 90% d-T₈₀-allethrin (Pynamin Forte 90EC, lot. 7304M901, Sumitomo Chemical Co., Ltd., Tokyo, Japan) was diluted with water to obtain a 250 ppm solution. After releasing the larva, we added 32 µL or 8 µL of the solution to each vial, to obtain a concentration of 0.4 ppm or 0.1 ppm, respectively. We used 10 larvae for each concentration regime, and observed larval knockdown for 30 min. Larvae that sank to the bottom of the glass vial and could not swim, float, or were paralyzed were judged as knocked down larva; we recorded the time to knockdown for each of these larva. We scored the median knockdown time (KT₅₀), i.e., the time required for 50% knockdown of larvae used for each concentration regime, according to the six following categories: 1, <5 min; 2, 5 to <10 min; 3, 10 to <15 min; 4, 15 to <20 min; 5, 20 to <30 min; and 6, ≥ 30 min. We calculated the susceptibility index as the product of the scores at 0.1 ppm and 0.4 ppm. Thus, mosquitoes with a susceptibility index of 1 were considered to be the most susceptible to d-allethrin, and those with a susceptibility index of 36 were considered to be the least susceptible to d-allethrin. After the bioassay, we identified each larva according to the keys provided by Rattanarithikul et al. [14]. The larvae were then stored in a 1.5-mL plastic vial containing ethanol solution, for subsequent polymerase chain reaction (PCR) analysis.

Analysis of the frequency of point mutations

To test for the presence of point mutations at \$989, I1011, L1014, V1016, and F1534 in Aedes aegypti, we used PCR and direct DNA sequencing. We performed the sequencing on 20 larvae/collection site, which were randomly selected after the bioassay. The mosquito sample (i.e., a larva immersed in ethanol) was lightly dried on a paper towel and placed in a 1.5-mL PCR reaction tube. The sample was homogenized in a mixed solution of extraction solution (40 μ L) plus tissue-preparation solution (10 μ L) (REDExtract-N-Amp Tissue PCR Kit; Sigma, St. Louis, MO) for extraction of DNA. The solution was heated at 95°C for 3 min and neutralized. Initial amplification was carried out using the primers AaSCF1 (AGACAATGTGGATCGCTTCC) and (GGACGCAATCTGGCTTGTTA) for S989P, AaSCR4 I1011M (or V), L1014F, and V1016G (or I); or AaSCF7 (GAGAACTCGCCGATGAACTT) and AaSCR7 (GACGAC-GAAATCGAACAGGT) for F1534C. The PCR mixture contained 4 µL of REDExtract-N-Amp ReadyMix (Sigma), 0.5 µM of each primer, and 1 µL of the DNA template in a total volume of 10 µL. PCR was performed under the following conditions: initial denaturation at 94°C for 3 min; 35 cycles each of 94°C for 15 s, 55°C for 30 s, and 72°C for 30 s; and a final elongation step at 72°C for 10 min. The amplified fragments of the expected size were purified with ExoSAP-IT (USB Corporation, Cleveland, OH) at 37°C for 30 min, and then 80°C for 15 min. DNA sequencing was carried out by using the primers AaSCF3 (GTGGAACTTCACCGACTTCA) and AaSCR6 (CGACTT-GATCCAGTTGGAGA) for S989P, I1011M (or V), L1014F; and V1016G (or I), or AaSCR8 (TAGCTTTCAGCGGCTTCTTC) for F1534C. A BigDye Terminator v 3.1 Cycle Sequencing Kit (Applied Biosystems Japan Ltd., Tokyo, Japan) was used for DNA sequencing, according to the manufacturer's instructions. Two micromoles of each primer were added to a tube, to give a total mixture volume of 10 µL. PCR was performed under the following conditions: initial denaturation at 96°C for 1 min; 25 cycles each of 96°C for 10 s, 50°C for 5 s, and 60°C for 2 min.

Statistical analysis

The susceptibility index and the corresponding frequency of point mutations for each collection point were plotted on a shape file map (DIVA-GIS, http://www.diva-gis.org/gdata) by using ArcGIS 10.2 (ESRI Japan Corp., Tokyo, Japan). An independent chi-squared test was performed for the number of larvae possessing V1016G and F1534C, or V1016G and S989P.

Results

Figure 1 shows the distribution pattern of *Aedes aegypti* and *Aedes albopictus* Skuse larvae collected from used tires. We collected 2,400 larvae, among which 2,332 were *Aedes aegypti*, 29 were *Aedes albopictus*, and 39 were other species, including *Culex quinquefasciatus* Say. We identified *Aedes aegypti* larvae in 55 of the 57 collection sites, indicating the predominance of this mosquito; by contrast, we identified *Aedes albopictus* larvae in only three sites. Figure 2 shows the distribution of the susceptibility

indices of *Aedes aegypti* larvae, as determined by the simplified knockdown bioassay. With the exception of a single location in North Dagon (susceptibility index, 30), the maximum susceptibility index (36) was recorded for *Aedes aegypti* larvae in 54 of the 55 sites where this species was collected. The percentages of knocked down larvae at 0.4 ppm of *d*-allethrin, determined 30 min after treatment, were 0% for 42 collection sites and 10–30% for the remaining 12 collection sites. The susceptibility index of *Aedes albopictus* collected at a single site in Dagon area was 18.

Figures 3 and 4 show the distribution of frequencies of point mutations at V1016, and F1534 in *Aedes aegypti* collected from used tires. We detected no mutations at I1011 and L1014 among 281 and 271 larvae sequenced, respectively. On the other hand, for V1016G, we detected 230 homozygous and 21 heterozygous mutations among 285 larvae sequenced (accession No. AB914689, AB914690). The frequency and % homozygous of the V1016G point mutation were 84.4% and 80.7%, respectively (Figure 3). For S989P, we detected 140 homozygous and 64 heterozygous point mutations among 218 larvae sequenced. The frequency and % homozygous of the S989P mutation were 78.8% and 64.2%, respectively. For the recently identified F1534C point mutation, we detected 16 homozygous and 105 heterozygous point mutations among 323 larvae sequenced (accession No. AB914687, AB914688). The frequency and % homozygous of the F1534 mutation were 21.2% and 5.0%, respectively (Figure 4).



Figure 1. Distribution of *Aedes aegypti* and *Aedes albopictus* larvae. The larvae were collected from used tires in Yangon City. Each circle indicates the species composition of the two species. doi:10.1371/journal.pntd.0003032.g001



Figure 2. Distribution of susceptibility indices in *Aedes aegypti* **larvae.** The susceptibility index was determined by using the simplified knockdown bioassay. The color in each circle indicates the susceptibility index. The larger the index, the lower the susceptible to *d*-allethrin. doi:10.1371/journal.pntd.0003032.g002

Although the number of sampling was few, no point mutation was detected in 20 larvae of *Ae. albopictus*.

We detected three patterns of co-occurrence of point mutations; V1016G/F1534C, V1016G/S989P, and V1016G/F1534C/S989P. We analyzed the haplotypes of each gene among 204 mosquitoes (Figure 5). Our results revealed that the percentage of larvae possessing the homozygous V1016G mutation and the wild type 1534F was significantly higher (59.3%) than the percentage of larvae possessing the homozygous V1016G and F1534C mutations (2.9%) (P<0.01, $\chi^2 = 72.6$). By contrast, the percentage of co-occurrence of the homozygous V1016G and S989P mutations was significantly high (65.7%) (P<0.01, $\chi^2 = 135$). Additionally, co-occurrences of the above three homozygous point mutations (V1016G/F1534C/S989P) were detected in 2 individuals among 204 individuals analyzed (0.98%).

Discussion

In the present study, we detected high frequencies of the V1016G (84.4%) and S989P (78.8%) mutations in *Aedes aegypti* collected from used tires in Yangon City. Additionally, we detected a relatively low allelic frequency of the F1534C mutation (21.2%). The I1011 and L1014 point mutations were undetected in *Aedes aegypti* in the study area.

High frequencies, namely, 3.1–61.8% [15] and 23% [16], of the V1016G mutation in *Aedes aegypti* were reported in Thailand. The V1016G mutation was also found in Vietnam; however, the

distribution was limited and the frequency of the point mutation was relatively low (2 heterozygotes were found among 10 larvae collected at a single site in Quang Ngai Province) [17]. At present, no other report of this point mutation in another Southeast Asian country has been published. The V1016I and I1011 (M or V) mutations seem to be widely distributed with high frequency in Latin American countries [7,11,18–21]. It is interesting to speculate why the V1016G point mutation occurs with high frequency in Myanmar and Thailand. Myanmar shares the longest border (2107 km) with Thailand among the neighboring countries, and therefore the two countries may share environmental, sociological, and biological similarities. Further studies to clarify the genetic relationships between two *Aedes aegypti* populations in Myanmar and Thailand, which may have had exchange or introgression of genes, are required.

After the first description of the newly identified F1534C point mutation in *Aedes aegypti* collected in Thailand [9,10], the same mutation was reported in Vietnam [17]; Brazil; Venezuela; Madeira Island, Portugal [22]; and Grand Cayman Island, UK [11]. In the present study, we determined a moderate frequency of this point mutation in Yangon City (21.2%); this frequency was of a comparable level to that recorded in Vietnam (average, 21.6%) [17], but lower than the frequencies recorded in Thailand (59–79%) [10], Grand Cayman Island (68%) [11], and Venezuela (100%) [22]. Additionally, the same mutation was recently reported in another DHF vector, *Aedes albopictus*, at high frequency (73%) [23]. Elucidation of the worldwide distribution



Figure 3. Distribution of the V1016G point mutation in *Aedes aegypti* larvae. The larvae were collected from used tires in Yangon City. Each circle indicates the allelic composition of the point mutations/wild types. doi:10.1371/journal.pntd.0003032.g003



Figure 4. Distribution of the F1534C point mutation in *Aedes aegypti* larvae. The larvae were collected from used tires in Yangon City. Each circle indicates the allelic composition of the point mutations/wild types. doi:10.1371/journal.pntd.0003032.g004



Figure 5. Genotyping of the V1016, F1534, and S989 point mutations in *Aedes aegypti* larvae. The larvae were collected from used tires in Yangon City. The total number of larval samples used for the analysis was 204. doi:10.1371/journal.pntd.0003032.q005

of the F1534C mutation in *Aedes aegypti* and *Aedes albopictus* will provide a valuable insight into DHF epidemiology, and yield useful information for vector control programs.

In the present study, we detected three patterns of cooccurrence of point mutations, namely, V1016G/S989P, V1016G/F1534C, and V1016G/F1534C/S989P. The S989P mutation has always been linked to the V1016G mutation; however, V1016G has sometimes been found in the absence of S989P [12]. The observed co-occurrence of the V1016G and S989P mutations with high frequency in Myanmar correspond to the former scenario. Srisawat et al. indicated that the cooccurrence of V1016G and S989P enhanced the resistance of Aedes aegypti to deltamethrin [12]. However, the authors were unable to determine the effect of the single S989P mutation on pyrethroid sensitivity, and were therefore unable to provide direct evidence of how the two mutations were associated with resistance. Recently, Du et al. reported that the co-occurrence of the V1016G and S989P mutations was not additive or synergistic to the V1016G mutation; however, the S989P mutation was concurrent with the V1016G mutation in a pyrethroid-resistant population of Aedes aegypti [24]. Further studies regarding the synergistic role of S989P on V1016G are required.

Interestingly, Stenhouse et al. [15] reported that, in an *Aedes aegypti* population in Thailand, wild type 1016V mosquitoes were homozygous for the F1534C mutation (43.5%); further, all 1016 heterozygous mosquitoes were 1534 heterozygous (38.8%), and homozygous V1016G mosquitoes were homozygous for the wild type 1534F (17.6%). These findings indicate that the wild type 1016V + homozygous F1534C, the homozygous V1016G, and the wild type 1534F exist with equal frequencies. On the contrary, in

the present study, we detected dominance of the homozygous V1016G + homozygous wild type 1534F (59.3%); further, the percentage of mosquitoes possessing the wild type 1016V + homozygous F1534C mutations was low (1.5%). Additionally, we detected a small number of mosquitoes possessing the homozygous V1016G + F1534C (2.9%) and V1016G + F1534C + S989P (0.98%). Recently, Linss et al. [25] detected the co-occurrence of V1016I and F1534C point mutations in Aedes aegypti populations in Brazil, and reported that the widespread distribution of the wild type 1016V + homozygous F1534C and the homozygous V1016I + F1534C in the country during the previous decade. Interestingly, the frequency of mosquitoes with the homozygous V1016I + F1534C mutation in Brazil (average 20.5%) was higher than the reported frequencies of mosquitoes with the homozygous V1016G + F1534C. In Thailand, the F1534C mutation is considerably more common than the V1016G mutation [10,15]. Mosquitoes with the homozygous F1534C mutation are thought to be susceptible to deltamethrin [15]. The I1011M and F1534C mutations seem to play a role in decreasing susceptibility to Type I pyrethroids, such as permethrin; on the other hand, the L1014F (or S) and V1016G mutation play important roles in the insensitivity of the voltage-gated sodium channel to Type II pyrethroids, such as deltamethrin [24,26]. Du et al. [24] indicated that the contribution of the V1016I mutation to pyrethroid resistance was lower than that of the V1016G mutation. Therefore, in the Aedes aegypti population investigated in the present study, the higher frequency of occurrence of the V1016G mutation than the F1534C mutation may indicate a history of pyrethroid treatment as part of vector control programs in the study sites. First, the use of DDT or Type I pyrethroids might select F1534C mutations, then V1016G has been

selected by the following use of type II pyrethroids. The possible increase of the homozygous V1016G and F1534C mutations associated with the S989P point mutation under increasing pyrethroid selection pressure may represent the worst case scenario for vector control programs.

The first outbreak of DHF was recorded in Myanmar in 1969. Since then, DHF has spread to 12 out of 14 states. During 1970-1991, 83381 cases, causing 3243 deaths were recorded [27]. The principal vector of DHF is Aedes aegypti, with Aedes albopictus acting as a secondary vector in suburban and rural areas. The growth of economic activities and traditional water storage practices increased the availability of vector breeding sites, thereby increasing the incidence of DHF cases from 5621 in 2005 to 11,049 in 2006 [28]. The Aedes mosquito control unit was established in 1968, and several feasibility studies aimed at the development of Aedes aegypti control methodology-including insecticide treatment, biological control by using larvivorous fish, environmental improvement, and health education-were undertaken [27]. However, information regarding insecticide usage for DHF vector control in Myanmar is lacking, and few studies concerning the insecticide resistance of DHF vectors have been published. Treatment of pyriproxyfen sand granules and BTI, and the use of these chemicals in combination with biological measures-such as the release of dragon fly nymphs [29]seemed to be acceptable to the residents of Yangon City [30]. During 1988-2005, 510 t of organochlorine insecticides, 360 kg of organophosphate insecticides, and 8157 kg of pyrethroid insecticides were used for malaria control in Myanmar. By contrast, only 1.2 t of organophosphate insecticides were used for DFH control during the same period. These data indicate that large amounts of organochlorine insecticides, including DDT, were used in Myanmar until the use of these insecticides was banned in 2003 [1]. The ban on organochlorine insecticides was introduced later in Myanmar (2003) and Thailand (2001) than in other Southeast Asian countries. Thaung et al. reported that DDT was routinely used for the control of the rat flea, Xenopsylla cheopsis (Rothschild), and that periodical monitoring of insecticide susceptibility in Yangon City indicated DDT resistance in rat fleas. The authors further suggested the existence of resistance to DDT, and lack of resistance to other types of insecticides, in Aedes aegypti [31].

In the present study, we determined high pyrethroid resistance in the larvae of *Aedes aegypti*. Interestingly, we recorded the maximum susceptibility index (36) in almost all of the collection sites. We also noted high DDT and pyrethroid (permethrin and deltamethrin) resistance in adult *Aedes aegypti* collected in the seven townships surveyed (unpublished data). Vu et al. [32] and Huber et al. [33] concluded that prolonged use of pyrethroids for malaria control may be one of the causative factors of pyrethroid resistance in *Aedes aegypti* in Vietnam. Since the abandonment of

References

- Zaim M, Jambulingam P (2007) Global insecticide use for vector-borne disease control, 3rd ed. World Health Organization.
- Kawada H (2009) An Inconvenient Truth of Pyrethroid– Does it have a promising future? In: Advances in Human Vector Control (ACS Symposium Book 1014). Clark J, Bloomquist JR, Kawada H, editors. New York: American Chemical Society.
- Martinez-Torres D, Chandre F, Williamson MS, Darret F, Bergé JB, et al. (1998) Molecular characterization of pyrethroid knockdown resistance (kdr) in the major malaria vector Anopheles gambiae s.s. Insect Mol Biol 7: 179– 184.
- Enayati AA, Vatandoost H, Ladonni H, Townson H, Hemingway J (2003) Molecular evidence for a kdr-like pyrethroid resistance mechanism in the malaria vector mosquito Anopheles stephensi. J Med Vet Entomol 17: 138–144.
- Martinez-Torres D, Chevillon C, Brun-Barale A, Bergé JB, Pasteur N, et al. (1999) Voltage-dependent Na⁺ channels in pyrethroid-resistant *Culex pipiens* L. mosquitoes. Pestic Sci 55: 1012–1020.

DDT sprays in Vietnam in 1995, a large number of photostable pyrethroids (λ -cyhalothrin, α -cypermethrin, deltamethrin, and permethrin) have been used to treat the interiors of houses as a residual spray, and to manufacture pyrethroid-impregnated bed nets for malaria control [1,34-36]. In Myanmar, pyrethroid insecticides were first used for malaria control in 1992, and have since been constantly used (>500 kg annually). Pyrethroid treatment for malaria vector control seems to have been intensively conducted in the interior and along the periphery of human habitation areas, where the breeding and resting sites of Aedes aegypti are located. This intensive usage may explain the strong selection pressure toward Aedes aegypti, because this mosquito is generally a domestic and endophagic species with a preference for indoor breeding [37–40]. Extensive use of DDT for malaria control before the use of this chemical was banned may also explain the development of pyrethroid resistance in Aedes aegypti, because the target site (i.e., the voltage-gated sodium channel) is common to DDT and pyrethroids.

Pyrethroids represent one of the most promising countermeasures for controlling malaria, DHF, and other arthropod-borne diseases. Currently, there are no suitable chemical substitutes for pyrethroids, and therefore pyrethroid resistance will be a major problem for vector control programs [41]. The development of new chemicals with novel modes of action, which can be substituted for conventional insecticides, is essential. Transitional life-prolonging measures for conventional 1st generation pyrethroids that belong to the photo-unstable knockdown agent groups and are used as "spatial repellents" [42], rotational use of plural insecticides with different mode of actions, and basic biochemical and genetic research to support the above strategies, are crucial to the effective management of insecticide resistance.

Acknowledgments

We thank U Thi Ha, Zar Zar Ag, Chit Thet New, Khin Myo Aye, Thu Zar Nyein Mu, and Yi Yi Myint, of the Medical Entomology Research Division, Department of Medical Research (Lower Myanmar), Yangon, Myanmar, for providing technical support, for rearing and providing the experimental insects, and for assisting with this study. We are grateful to L. Yoshida, Institute of Tropical Medicine, Nagasaki University, for giving us facilities for the onset of the study.

Author Contributions

Conceived and designed the experiments: HK SZMO KZT NM. Performed the experiments: HK SZMO ST EK YNMM HMT. Analyzed the data: HK SZMO ST EK. Contributed reagents/materials/analysis tools: YNMM HMT. Wrote the paper: HK. Provided critical comment on the manuscript and final approval of the version to be published: KZT NM.

- Brengues C, Hawkes NJ, Chandre F, McCaroll L, Duchon S, et al. (2003) Pyrethroid and DDT cross-resistance in *Aedes aegypti* is correlated with novel mutations in the voltage-gated sodium channel gene. Med Vet Entomol 17: 87– 94.
- Saavedra-Rodriguez K, Urdaneta-Marquez L, Rajatileka S, Moulton M, Flores AE, et al. (2007) A mutation in the voltage-gated sodium channel gene associated with pyrethroid resistance in Latin American *Aedes aegypti*. Insect Mol Biol 16: 785–798.
- Chang C, Shen W-K, Wang T-T, Lin Y-H, Hsu E-L, et al. (2009) A novel amino acid substitution in a voltage-gated sodium channel is associated with knockdown resistance to permethrin in *Aedes aegypti*. Insect Biochem Mol Biol 39: 272–278.
- Yanola J, Somboon P, Prapanthadara L (2008) A novel point mutation in the *Aedes aegypti* voltage-gated sodium channel gene associated with permethrin resistance. The 2nd International Conference on Dengue and Dengue Haemorhagic Fever, Oct. 15–17, 2008, Phuket, Thailand.

- Yanola J, Somboon P, Walton C, Nachaiwieng W, Somwang P, et al. (2011) High-throughput assays for detection of the F1534C mutation in the voltagegated sodium channel gene in permethrin-resistant *Aedes aegypti* and the distribution of this mutation throughout Thailand. Trop Med Int Health 16: 501–9.
- Harris AF, Rajatileka S, Ranson H (2010) Pyrethroid resistance in Aedes aegypti from Grand Cayman. Am J Trop Med Hyg 83: 277–284.
- Srisawat R, Komalamisra N, Eshita Y, Zheng M, Ono K, et al. (2010) Point mutations in domain II of the voltage-gated sodium channel gene in deltamethrin-resistant *Aedes aegypti* (Diptera: Culicidae). Appl Entomol Zool 45: 275–282.
- Kawada H, Higa Y, Nguyen YT, Tran SH, Nguyen HT, et al. (2009) Nationwide investigation of the pyrethroid susceptibility of mosquito larvae collected from used tires in Vietnam. PLoS Negl Trop Dis 3: e391.
- Rattanarithikul R, Harbach RE, Harrison BA, Panthusiri P, Coleman RE, et al. (2010) Illustrated keys to the mosquitoes of Thailand. VI. Tribe Aedini. Southeast Asian J Trop Med Public Health 41 Suppl 1: 1–225.
- 15. Stenhouse SA, Plernsub S, Yanola J, Lumjuan N, Dantrakool A, et al. (2013) Detection of the V1016G mutation in the voltage-gated sodium channel gene of *Aedes aegypti* (Diptera: Culicidae) by allele-specific PCR assay, and its distribution and effect on deltamethrin resistance in Thailand. Parasit Vectors 6: 253.
- Rajatileka S, Black IV WC, Saavedra-Rodriguez K, Trongtokit Y, Apiwathnasorn C, et al. (2008) Development and application of a simple colorimetric assay reveals widespread distribution of sodium channel mutations in Thai population of *Aedes aegypti*. Acta Trop 108: 54–57.
- Kawada H, Higa Y, Komagata O, Kasai S, Tomita T, et al. (2009) Widespread distribution of a newly found point mutation in voltage-gated sodium channel in pyrethroid-resistant *Aedes aegypti* populations in Vietnam. PLoS Negl Trop Dis. 3: e527.
- Siller Q, Ponce G, Lozano S, Flores AE (2011) Update on the frequency of Ile1016 mutation in voltage-gated sodium channel gene of *Aedes aegypti* in Mexico. J Am Mosq Control Assoc 27: 357–362.
- Martins AJ, Lima JBP, Peixoto AA, Valle D (2009) Frequency of Val1016Ile mutation in the voltage-gated sodium channel gene of *Aedes aegypti* Brazilian populations. Trop Med Int Health 14: 1351–1355.
- McAllister JC, Godsey MS, Scott ML (2012) Pyrethroid resistance in Aedes aegypti and Aedes albopictus from Port-au-Prince, Haiti. J Vect Ecol 37: 325– 332.
- Marcombe S, Mathieu RB, Pocquet N, Riaz MA, Poupardin R, et al. (2012) Insecticide resistance in the dengue vector *Aedes aegypti* from Martinique: Distribution, mechanisms and relations with environmental factors. PLoS One 7: e30989.
- Seixas G, Salgueiro P, Silva AC, Campos M, Spenassatto C, et al. (2013) Aedes aegypti on Madeira Island (Portugal): genetic variation of a recently introduced dengue vector. Mem Inst Oswaldo Cruz 108 Suppl 1: 3–10.
- Kasai S, Ng LC, Lam-Phua SG, Tang CS, Itokawa K, et al. (2011) First detection of a putative knockdown resistance gene in major mosquito vector, *Aedes albopictus*. Jpn J Infect Dis 64: 217–21.
- Du Y, Nomura Y, Satara G, Hua Z, Nauenb R, et al. (2013) Molecular evidence for dual pyrethroid-receptor sites on a mosquito sodium channel. PNAS 110: 11785–11790.

- Linss JGB, Brito LP, Garcial GA, Araki AS, Bruno RV, et al. (2014) Distribution and dissemination of the Val1016Ile and Phe1534Cys Kdr mutations in Aedes aegypti Brazilian natural populations. Parasit Vect 7: 25.
- Hu Z, Du Y, Nomura Y, Dong K (2011) A sodium channel mutation identified in *Aedes aegypti* selectively reduces cockroach sodium channel sensitivity to type I, but not type II pyrethroids. Insect Biochem Mol Biol 41: 9–13.
- Aung TT, Win S, Aung S (1996) Status report on epidemiology of dengue/ dengue haemorrhagic fever in Myanmar, 1995. Dengue Bull 20: 41–45.
- Oo TT, Storch V, Madon MB, Becker N (2011) Factors influencing the seasonal abundance of *Aedes (Stegomyia) aegypti* and the control strategy of dengue and dengue haemorrhagic fever in Thanlyin Township, Yangon City, Myanmar. Trop Biomed 28: 302–311.
- Sebastian A, Thu MM, Kyaw M, Sein MM (1980) The use of dragonfly nymphs in the control of *Aedes aegypti*. Southeast Asian J Trop Med Public Health 11: 104–7.
- Wai KT, Htun PT, Oo T, Myint H, Lin Z, et al. (2012) Community-centred eco-bio-social approach to control dengue vectors: an intervention study from Myanmar. Pathog Glob Health 106: 461–8.
- Thaung U, Ming CK, Thein M (1975) Insecticide susceptibility of some vector fleas and mosquitoes in Burma. Southeast Asian J Trop Med Public Health 6: 555–61.
- Vu DH, Nguyen TBN, Do TH, Nguyen TBL (2004) Susceptibility of Aedes aegypti to insecticides in Viet Nam. Dengue Bull 28: 179–183.
- Huber K, Luu LL, Tran HH, Tran KT, Rodhain F, et al. (2003) Aedes aegypti in south Vietnam: Ecology, genetic structure, vectorial competence and resistance to insecticides. South East Asian J Trop Med Public Health 34: 81–86.
- Verlé P, Lieu TTT, Kongs A, der Stuyft PV, Coosemans M (1999) Control of malaria vectors: cost analysis in a province of northern Vietnam. Trop Med Int Hlth 4: 139–145.
- Hung LQ, de Vries PJ, Giao PT, Nam VS, Binh TQ, et al. (2002) Control of malaria: a successful experience from Viet Nam. Bull WHO 80: 660–666.
- Nam NV, de Vries PJ, Toi LV, Nagelkerke N (2005) Malaria control in Vietnam: the Binh Thuan experience. Trop Med Int Hlth 10: 357–365.
- Hawley WA (1988) The biology of *Aedes albopictus*. J Am Mosq Control Assoc (Suppl. 1) 4: 1–39.
- Edman JD, Kittayapong P, Linthicum K, Scott T (1997) Attractant resting boxes for rapid collection and surveillance of *Aedes aegypti* (L.) inside houses. J Am Mosq Control Assoc 13: 24–27.
- Ishak H, Miyagi I, Toma T, Kamimura K (1997) Breeding habitats of Aedes aegypti (L.) and Aedes albopictus (Skuse) in villages of Barru, South Sulawesi, Indonesia. Southeast Asian J Trop Med Public Health 28: 844–850.
- 40. Tsuda Y, Suwonkerd W, Chawprom S, Prajakwong S, Takagi M (2006) Different spatial distribution of *Aedes aegypti* and *Aedes albopictus* along an urban-rural gradient and the relating environmental factors examined in three villages in northern Thailand. J Am Mosq Control Assoc 22: 222–228.
- 41. Nauen R (2007) Insecticide resistance in disease vectors of public health importance. Pest Manag Sci 63: 628–633.
- Kawada H (2012) New mosquito control techniques as countermeasures against insecticide resistance. In: Insecticides - Advances in Integrated Pest Management. Perveen F, editor. InTech. 657–682 p.